**Author's response to reviews**

**Title:** Women's position and attitudes towards Female Genital Mutilation in Egypt. Results from the Demographic and Health Surveys, 1995-2008

**Authors:**

Ronan Van Rossem (ronan.vanrossem@ugent.be)
Dominique Meekers (dmeekers@tulane.edu)
Anastasia Gage (agage@tulane.edu)

**Version:** 2  **Date:** 12 May 2015

**Author's response to reviews:**

- line numbers added
- Abbreviations, Competing interest, Author's contribution and Acknowledgement sections moved after the “Conclusion” section
- FGM spelled out in title.